Author’s response to reviews

Title: Prevalence of Unwanted Pregnancy among Iranian Women: An Updated Meta-Analysis

Authors:
Kourosh Sayehmiri (Sayehmiri@razi.tums.ac.ir)
Fariba Ebtekar (Ebtekarf@gmail.com)
Mozhdeh Zarei (Mozh.zarei@yahoo.com)
Reza Ghanei Gheshlagh (rezaghanei30@yahoo.com)

Version: 2 Date: 07 Nov 2019

Author’s response to reviews:

We sincerely thank the reviewers for dedicating their time to review our meta-analysis. We assess the reviewers' comment.

1- Since the method section of the article overlaps with the other two articles by the same author, those sections were re-written and all changes were highlighted in green.

2- Table 2 was removed from the text due to overlap with other articles and its description was added to the text.